Author’s response to reviews

Title: A Rare Cause Of Abnormal Uterine Bleeding

Authors:

Mridula A Benjamin (mridulaben@yahoo.com)
Hjh Roselina Yaakub (yaakubroselina@hotmail.com)
PU Telesinghe (ptelsinghe@yahoo.com.sg)
Gazala Kafeel (gbilgrami@hotmail.com)

Version: 3 Date: 2 February 2010

Author’s response to reviews:

From
Dr. Mridula A Benjamin
Dep of Obstetrics and Gynaecology
RIPAS Hospital
Bandar Seri Begawan
Brunei
To
Independent Editorial Production Team
BioMed Central
Sir/madam,
As requested by you I have made the following changes.
1. Included the ethnicity and nationality of the patient
2. Removed the figures from the manuscript
3. Have given the legend and title of the figures after the references.
4. Uploaded the figure files separately
5. Included citations for the figures
6. Have mentioned that all the authors have read the manuscript and approved
7. Have changed the way the references appear into consecutive numbers in square brackets before the full stop.
8. Removed all underlines
9. Changed the title to include “case report” and all into sentence case.
10. Removed the heading “authors”
11. Authors qualifications and job title was not mentioned. Only work place is mentioned.
12. E-mail addresses have been added
13. Consent, Competing Interests, and Authors' Contributions sections to before the reference list.
14. Additional file has been removed
15. Have checked the manuscript for any typographical errors

I'm not sure whether the figure has been mentioned in the proper way in the manuscript. Also I have attached the figures as JPEG format. If any changes have to be made, kindly let me know or the necessary changes may be made.

Thanking you and with regards
Dr. Mridula a Benjamin.